Author’s response to reviews

Title: Fulminant encephalopathy in a child with hyperferritinemic sepsis: a case report

Authors:

Lilin Huang (921295597@qq.com)
Shumei Peng (pengsm2010@163.com)
Ronghan Li (150348200@qq.com)
Danyu Xie (13580362856@163.com)
Dongping Huang (hdpaaa@126.com)

Version: 5 Date: 22 Feb 2020

Author’s response to reviews:

Point-by-point responses to assistant editor:

Assistant Editor Comments:

A. Authors' Contributions
Please represent authors' names using their full initials, not their full name, in the Authors’ Contributions section. If there are any duplicated initials, please differentiate them to make it clear that the initials refer to separate authors.

Response: We are really sorry for the mistake. The authors' names have been amended using their full initials in line 276-278, page 10 in Authors’ contributions section which is described as below:
“Study concept and design: LH. Acquisition of data: RL and DX. Analysis and interpretation of data: LH and DH. Drafting of the manuscript: LH, SP and DH. All authors have read and approved the manuscript.”